Author's response to reviews

Title: The high affinity selectin glycan ligand C2-O-sLex and mRNA transcripts of the core 2 beta-1,6-N-acetylglusaminyltransferase (C2GnT1) gene are highly expressed in human colorectal adenocarcinomas

Authors:

  Catherine A. St. Hill (sthil001@umn.edu)
  Mariya Farooqui (faroo001@umn.edu)
  Gregory Mitcheltree (mitc0120@umn.edu)
  H. Evin Gulbahce (gulba001@umn.edu)
  Jose Jessurun (jessu001@umn.edu)
  Qing Cao (caox0075@umn.edu)
  Bruce Walcheck (walch003@umn.edu)

Version: 4 Date: 4 March 2009

Author's response to reviews: see over
March 4, 2009

Melissa Norton, M.D.
Editor-In-Chief
BMC Cancer
Biomed Central Ltd.
Middlesex House
34-42 Cleveland Street
London, W1T 4LB, UK

RE: MS: 4306465602167484

Dear Dr. Norton,

We have revised our manuscript titled, “The high affinity selectin glycan ligand C2-O-sLe\textsuperscript{x} and mRNA transcripts of the core 2 \(\beta\)-1,6-\(\text{N}\)-acetylglusaminyltransferase (C2GnT1) gene are highly expressed in human colorectal adenocarcinomas” according to the formatting requirements. A point-by-point description of the changes made is outlined below:

1. The Reference section was modified to provide a complete list of all authors and the volume/issue/page numbers for all references.

2. The figure title and number were removed from each image file. The figure citations in the body of the manuscript were changed so that they are in consecutive order and all figures were cited. The text was modified as necessary to reflect these changes. Ranges were not used when listing the figures.

3. Line numbers were removed from the text.

4. The figures were cropped to minimize white space around the images.

5. The manuscript was checked for errors and corrections were made as necessary.

Thank you for your review of this manuscript. I can be reached preferably by e-mail at sthil001@umn.edu; at the mailing address above; by telephone at 612-626-5787; or by fax at 612-624-0751 or with any questions or concerns.

Sincerely,

Catherine St. Hill, D.V.M., Ph.D